Author’s response to reviews

Title: Teledentistry as a Novel Pathway to Improve Dental Health in School Children: A Research Protocol for a Randomised Controlled Trial

Authors:

Mohamed Estai (abdalla177@gmail.com)
Yogesan Kanagasingam (yogi.kanagasingam@csiro.au)
Maryam Mahdizadeh (maryam.mehdizadeh@csiro.au)
Janardhan Vignarajan (janardhan.vignarajan@csiro.au)
Richard Norman (richard.norman@curtin.edu.au)
Boyen Huang (bhuang@csu.edu.au)
Heiko Spallek (heiko.spallek@sydney.edu.au)
Michelle Irving (michelle.irving@sydney.edu.au)
Amit Arora (a.arora@westernsydney.edu.au)
Estie Kruger (estie.kruger@uwa.edu.au)
Marc Tennant (marc.tennant@uwa.edu.au)

Version: 2 Date: 13 Dec 2019

Author’s response to reviews:

Dear Editor

I would like to thank for your suggestions and considering our paper for publication in BMC Oral Health. A point-by-point response as following:

1. Ethics approval and consent to participate: Please alter your statement on consent in this section to clarify that parental consent will be obtained for the minors in the study.

Suggested statement has been added to Ethics approval and consent to participate section.
All comments have been incorporated in the main text. If you need further information, please let us know.

Sincerely yours

Thank you